Author's response to reviews

Title: Posterior laryngitis: a disease with different aetiologies affecting health-related quality of life. A prospective case-control study

Authors:

Hillevi Pendleton (Hillevi@Pendleton.se)
Marianne Ahlner-Elmqvist (Marianne.Ahlner-Elmqvist@med.lu.se)
Rolf Olsson (Rolf.Olsson@med.lu.se)
Ola Thorsson (Ola.Thorsson@med.lu.se)
Oskar Hammar (Oskar.Hammar@med.lu.se)
Magnus Jannert (Magnus.Jannert@med.lu.se)
Bodil Ohlsson (Bodil.Ohlsson@med.lu.se)

Version: 2 Date: 25 August 2013

Author's response to reviews: see over
Dear Dr Christopher Foote

Enclosed you will find the revised manuscript entitled “Posterior laryngitis: a disease with different aetiologies affecting health-related quality of life. A prospective case-control study”.

As recommended by reviewer Petros Vlastarakos, we have now revised in the Conclusion of the Abstract of the manuscript the sentence “This study indicates that a minority of patients with PL has LPR and is not cured by acid-reducing therapy”, to “This study indicates that a minority of patients with PL has LPR and is cured by acid-reducing therapy”

Yours sincerely,

Hillevi Pendleton